Research Paper





Restriction of the Rights and Coercion of Patients in a Psychiatric Hospital: The Opinion of People With Mental Disorders and Psychiatrists

Natalia Rzhevskaya1* (D)

1. Department of Psychiatry, Narcology, and Clinical Psychology, Belgorod State National Research University, Belgorod, Russia.



Citation Rzhevskaya N. Restriction of the Rights and Coercion of Patients in a Psychiatric Hospital: The Opinion of People With Mental Disorders and Psychiatrists. Iranian Rehabilitation Journal. 2023; 21(3):533-542. http://dx.doi.org/10.32598/irj.21.3.1913.1





Article info:

Received: 16 Oct 2022 Accepted: 29 May 2023 Available Online: 01 Sep 2023

Keywords:

Psychiatric care, Coercion, Physical restraint, Patients' rights, Bioethics

ABSTRACT

Objectives: Restriction of the rights of patients in a psychiatric hospital, isolation and fixation, compulsory treatment, and round-the-clock monitoring are negatively perceived by them, contribute to the stigma of a psychiatric hospital, and prevent timely access to psychiatric help We assessed the opinions of patients in psychiatric hospitals and psychiatrists about coercion and violence in the provision of psychiatric care for recommendations on their prevention.

Methods: An anonymous survey of psychiatrists and patients was conducted in psychiatric hospitals in three regions. Data were analyzed using descriptive and non-parametric statistical methods.

Results: Psychiatrists and patients were positive about the coercive measures in the psychiatric hospital. Physical restraint was considered the prerogative of orderlies by 64.5% of psychiatrists and 35.4% of patients. According to 19.6% of doctors and 28.4% of patients, a psychiatrist can independently apply physical restraint to aggressive patients. Injections of sedative drugs by a psychiatrist personally were considered justified by 81.3% of physicians and 64.6% of patients. Most patients and physicians noted that the psychiatrist needed to be involved in the application of physical restraint to an aggressive patient.

Discussion: Psychiatry is currently dominated by an archaic paternalistic model of doctorpatient relations and the delegation of additional functions of social control to psychiatrists. The introduction of a contractual model is required for more effective interaction between the psychiatrist and the patient.

Natalia Rzhevskaya, MD.

Address: Department of Psychiatry, Narcology, and Clinical Psychology, Belgorod State National Research University, Belgorod, Russia. Tel: +7 (910) 3214585

E-mail: konst_v@list.ru

^{*} Corresponding Author:

Highlights

- Anosognosia in psychiatric patients, combined with self-stigmatization, prevents the formation of sustainable compliance and responsibility for one's own health.
- The paternalistic model of the relationship between the psychiatrist and the patient places the responsibility for treatment and social functioning on the doctor, which can lead to abuse of the mentally ill patient.
- Prevention of coercion and violence in psychiatric hospitals should be based on the contractual model of the relationship between the doctor and the patient, psychoeducation, and regulations for the application of restrictive measures.

Plain Language Summary

In psychiatric hospitals, restrictions on the rights and freedoms of people with mental disorders are often allowed. This study examined the attitudes of patients and psychiatrists toward restrictive measures in psychiatric hospitals. It has been established that the majority of psychiatrists and patients have a positive attitude toward coercion in a psychiatric hospital, which indicates the predominance of a paternalistic model of assistance in psychiatry. It is necessary to move to a contractual model of psychiatrist-patient relationships and increase the responsibility of psychiatric patients for their health.

Introduction

he legislation of most countries establishes the principles of protecting the rights of people with mental disorders as a socially unprotected contingent [1]. An open information space helps to develop legal literacy and awareness of mental health care recipients, which increases the quality of medical care and respect for their rights [2]. Therefore, changes in psychiatric services should be comprehensive [3]. The stigma of the "mentally ill" in the public mind leads to discrimination against psychiatric patients [4], reducing their social status and quality of life [5].

Mental health workers treat the rights and problems of their patients with disdain [6]. In a psychiatric hospital, the rights of patients are often violated: They are forced to be hospitalized, they are denied access to medical documents, they are monitored around the clock, they control their medication, and they are prevented from refusing treatment [7]. Involuntary hospitalization becomes an obstacle to compromise between the timeliness of medical care, public safety, and the patient's right to liberty [8]. Restrictions on freedom of movement, physical restraints, and forced prescription of drugs used in psychiatric hospitals are perceived negatively by patients [9] and abused by medical staff [10].

The absence of laws on the powers of medical personnel to restrict the rights of patients, and the low level of legal literacy and knowledge in the field of bioethics of psychiatrists, paramedical personnel, and paramedical personnel lead to discrimination against psychiatric patients [11]. Coercion reduces the level of compliance of patients [12] and contributes to an increase in the frequency and severity of exacerbations of mental disorders [13].

Therefore, we assessed the opinions of people with mental disorders and psychiatrists about violence and coercion in a psychiatric hospital in order to prepare recommendations for their prevention.

Materials and Methods

The research was conducted in psychiatric hospitals in three regions of the Russian Federation-Belgorod, Volgograd, and Voronezh regions, in 2019. The objects of the study were divided into two groups: Psychiatric patients and psychiatrists.

Inclusion criteria for patients: The age of 18 years or older, diagnosis of F0 organic mental disorder, F2 schizophrenia, or F3 mood disorder established in accordance with the ICD-10 criteria, treatment in the acute psychiatric department of a psychiatric hospital, the state of becoming a therapeutic remission or persistent improvement in mental state. The exclusion criteria were the age of less than 18 years, acute psychopathological

symptoms, and severe cognitive and emotional-volitional deficit.

The inclusion criterion for the group of psychiatrists was work experience in an inpatient psychiatric department for at least one year and their exclusion criterion was work experience in an inpatient psychiatric unit for less than 12 months.

An anonymous survey of 107 psychiatrists and 271 patients was conducted using a medical-sociological and psychometric method using the author's questionnaires. The questionnaire for psychiatrists contained the following blocks: Socio-demographic information, opinions about coercion and violence in a psychiatric hospital, and assessment of one's own social distance from people with mental disorders. The questionnaire for patients contained the following blocks: Socio-demographic data, opinions about coercion and restrictive measures in a psychiatric hospital, and criticism of the disease.

Bogardus social distance scale [14] measures social distance as the degree of closeness or alienation between people with mental disorders and respondents. The average score was used to determine the social distance of the respondents in their attitude toward psychiatric patients. Five variants of social distance were distinguished: Close relations (points 1-2), open relations (points 3-4), distancing (point 5), isolation (point 6), and rejection (point 7).

The database was processed using nonparametric statistic methods (descriptive statistics, chi-square with Yates correction for 2x2 contingency tables, and odds ratio) using the Statistica software, version 8.

Results

The group of patients included 271 people: 110 men aged 42.3±11.5 years and 161 women aged 41.8±13.4 years. Among the patients, those with schizophrenic spectrum disorders predominated (78.2%), followed by those with affective disorders (11.4%), and those with organic, including symptomatic, and mental disorders (10.4%).

The group of psychiatrists included 107 people: 42 men aged 37.8±13.7 years and 65 women aged 34.2±11.2 years. Work experience in the "psychiatry" specialty was 10.6±10.7 years. Also, 40.1% of psychiatrists had the highest and first qualification category.

In the group of patients, 50.6% had persistent severe mental disorders and were disabled. Among the remaining psychiatric patients, more than half (61.2%) did not work.

In addition, 23.6% of patients completely denied the existence of a mental disorder and 49.5% of patients noted various psychological problems. A mild mental disorder was recognized in 16.6% of both sexes. Full-fledged criticism of their mental disorder was noted only in 10.3% of patients.

Also, 49.5% of psychiatrists considered themselves completely healthy and 44% of psychiatrists had psychological problems, while in 5% of the respondents, these problems led to a violation of social functioning. In addition, 6.5% of physicians indicated that they were being treated by a psychiatrist or psychotherapist.

Also, 35.8% of patients and 45.8% of psychiatrists consider it justified to force psychiatrists to treat people with mental disorders in any situation, the same number of respondents -30.6% and 40.2%, respectively - in cases of auto- and hetero-aggression (Table 1). In addition, 91.6% of psychiatrists and 72.3% of patients had a positive attitude toward the use of coercion in a psychiatric hospital (χ^2 =15.376, P=0.000, OR=4.1, and 95% CI, 1.9%-9.3%) and 7.4% of patients and 5.6% of psychiatrists spoke about the inadmissibility of the use of coercion and violence by psychiatrists under any circumstances. Coercion by nurses in all cases was considered justified by 23.6% of patients and 24.3% of psychiatrists. A similar number of respondents in both groups considered it possible to coerce into treatment only in the case of the patient's aggression (28.4% of patients and 29% of psychiatrists). Also, 26.2% of psychiatrists and 12.2% of patients did not allow the use of coercion and violence by nurses in the treatment ($\chi^2=10.087$, P=0.002, OR=2.6, and 95% CI, 1.4%-4.7%).

Regarding the opinion on the use of physical restraints during voluntary hospitalization by a psychiatrist personally with agitation in a patient (Table 2), 90.7% of psychiatrists and 74.9% of patients reported allowing their use (χ^2 =10.6732, P=0.0019, OR=3.2, and 95%CI, 1.5%-7.0%). The implementation of physical constraint was considered the prerogative of orderlies by 64.5% of psychiatrists (χ^2 =25.173, P=0.0005, OR=3.3, and 95% CI, 3.2%-5.4%) and 35.4% of patients. If a patient develops agitation, according to 28.4% of patients and 19.6% of psychiatrists, the psychiatrist can apply a physical constraint on his/her own. Also, 11.1% of patients and

Table 1. Opinion on the coercion of psychiatric patients to treatment by psychiatrists and nurses

	- 1: 10: ff —	No. (%)	
Coercion is Yustified	Medical Staff	Patients	Psychiatrists
· 16 % 1	Psychiatrists applied	97(35.8)	49(45.8)
Yes, definitely	Applied by nurses	64(23.6)	26(24.3)
	Psychiatrists applied	83(30.6)	43(40.2)
Only in case of aggression	Applied by nurses	77(28.4)	31(29.0)
	Psychiatrists applied	16(5.9)	6(5.6)
When treatment is refused	Applied by nurses	30(11.1)	7(6.5)
N	Psychiatrists applied	20(7.4)	6(5.6)
Not justified in any way	Applied by nurses	33(12.2)	28(26.2)
Difficult to answer	Psychiatrists applied	55(20.3)	3(2.8)
	Applied by nurses	67(24.7)	15(14)
	Psychiatrists applied	271(100)	107(100)
Total	Applied by nurses	271(100)	107(100)

∏ranian R∈habilitation **J**ournal

6.5% of psychiatrists delegated the physical restraint of people with mental disorders to nurses.

The use of injections of sedative drugs by a psychiatrist personally in the event of a patient developing psychomotor agitation was considered justified by 81.3% of psychiatrists and 64.6% of patients (χ^2 =9.3268, P=0.0032, OR=2.3, and 95% CI, 1.3%-4.2%) (Table 3).

Also, 66.4% of patients and 46.7% of psychiatrists allowed the involvement of other patients to help administer sedatives (χ^2 =11.674, P=0.001, OR=2.3, and 95%

CI, 1.4%-3.7%). In addition, 23.3% of psychiatrists approved the involvement of other patients in holding an aggressive patient in exceptional cases.

Among the respondents, 71.9% of patients and 72.9% of psychiatrists approved of the presence of special restraint belts for the application of physical restraint. Also, 59.8% of patients and 50.5% of psychiatrists spoke about the expediency of having "straitjackets" in psychiatric hospitals.

Table 2. Opinion on the use of physical restraint in case of arousal by a psychiatrist personally during voluntary hospitalization

Group Applicability	No	. (%)
	ty Patients	Psychiatrists
Yes, definitely	77(28.4)	21(19.6)
This should be done by nurses	30(11.1)	7(6.5)
This should be done by orderlies	96(35.4)	69(64.5)
Cannot be applied	68(25.1)	10(9.3)
Total	271(100)	107(100)

∏ranian Rehabilitation Journal

Table 3. Opinion about enlisting other patients to assist in administering sedative injections to aggressive patients

Group	No. (%)	
Recruiting Other Patients	Patients	Psychiatrists
Yes, it's acceptable	51(18.8)	9(8.4)
Yes, if he/she wants to help	52(19.2)	6(5.6)
Yes, if the medical staff is threatened	40(14.8)	10(9.3)
Yes, in exceptional cases	37(13.7)	25(23.4)
Cannot be attracted	91(33.6)	57(53.3)
Total	271(100)	107(100)

∏ranian Rehabilitation Dournal

In addition, 76.4% of patients and 82.2% of physicians (Table 4) expressed that psychiatrists could personally participate in the application of physical restraint in case of agitation, which threatens the lives of others and the absence of nursing staff nearby. The inadmissibility of the participation of a psychiatrist in the implementation of measures of physical restraint was indicated by 23.6% of patients and 17.8% of psychiatrists. The use of physical constraint by nurses on their own initiative, without the appointment of a psychiatrist, was considered acceptable by the majority of respondents in the case of psychomotor agitation in a patient with heteroaggression. Also, 88.8% of psychiatrists and 78.2% of patients admitted the participation of nurses in the application of physical restraint (χ^2 =4.933, P=0.027, OR=2.2, and 95% CI, 1.1%-4.5%).

The need for compulsory treatment of psychiatric patients (Table 5) "for the benefit of society" was admitted by 29.9% of patients and 10.3% of psychiatrists (χ^2 =14.971 P=0.0007 OR=3.7, and 95% CI, 1.8%-7.8%). The need for involuntary treatment of patients with severe mental disorders "in their own favor" was noted by 24.4% of patients and 16.8% of psychiatrists. Public danger as the only reason for compulsory treatment was indicated by 67.3% of psychiatrists and 31.7% of patients (χ^2 =38.4159, P=0.0005, OR=4.4, and 95% CI, 2.7%-7.4%). Also, 14.0% of patients and 5.6% of psychiatrists (χ^2 =4.4950, P=0.003, OR=2.7, and 95% CI, 1.1%-7.5%) considered the use of involuntary treatment of psychiatric patients unacceptable.

Table 4. Opinion on the need for participation in the application of physical restraint personally by a psychiatrist and nurses

Participation in the Application	Medical Staff	No. (%)	
of Physical Restraint		Patients	Psychiatrists
Yes, if other people's lives are in danger	Psychiatrist	127(46.9)	60(56.0)
	Nurse	120(44.3)	48(44.9)
V	Psychiatrist	80(29.5)	28(26.2)
Yes, in the absence of orderlies	Nurse	92(33.9)	47(43.9)
No involvement	Psychiatrist	64(23.6)	19(17.8)
no involvement	Nurse	59(21.8)	12(11.2)
Total	Psychiatrist	271(100)	107(100)
	Nurse	271(100)	107(100)

Iranian Rehabilitation Journal

Table 5. Opinion on the need for coercion in the treatment of patients with severe mental disorders

Group	No. (%)	
The Need for Treatment	Patients	Psychiatrists
Yes, for the benefit of society	81(29.9)	11(10.3)
Yes, for the benefit of the patient	66(24.4)	18(16.8)
Yes, in case of danger to themselves and others	86(31.7)	72(67.3)
No, not justified without their desire	38(14.0)	6(5.6)
Total	271(100)	107(100)

■ranian Rehabilitation Journal

As for the mandatory control of medication intake by patients in a psychiatric hospital by nurses (Table 6), 73.1% of patients and 89.7% of psychiatrists considered it mandatory. According to 62.6% of psychiatrists and 35.5% of patients, nurses must necessarily control the intake of medications, since patients often do not

take, but hide or throw away the dispensed medications (χ^2 =22.0319, P=0.001 OR=3.1, and 95% CI, 1.8%-5.0%). In addition, 26.9% of patients and 10.3% of psychiatrists (χ^2 =11.3698, P=0.002, OR=3.2, and 95% CI, 1.6%-6.7%) held opinions about the inappropriateness of drug control.

Table 6. Opinions on the need to control the intake of medicines by nurses

Group	N	No. (%)
Control is Necessary	Patients	Psychiatrists
Yes, because the patients hide the drugs	96(35.5)	67(62.6)
Yes, definitely	102(37.6)	29(27.1)
No control required	73(26.9)	11(10.3)
Total	271(100)	107(100)

∏ranian R∈habilitation Dournal

Table 7. Social distance of patients and psychiatrists in relation to people with severe mental disorders

Group	No. (%)		
Social Distance	Patients	Psychiatrists	
Acceptance as close relatives, marriage	51(18.8)	2(1.9)	
Acceptance as friends	59(21.8)	10(9.3)	
Acceptance as neighbors living on my street	47(17.3)	28(26.2)	
Acceptance as residents of my city	16(5.9)	5(4.7)	
Acceptance as citizens of my country	69(25.5)	55(51.4)	
Acceptance only as tourists in my country	3(1.1)	1(0.9)	
Would prefer not to see them in my country	26(9.6)	6(5.6)	
Total	271(100)	107(100)	

∏ranian **R**∈habilitation **J**ournal

The use of forced feeding was supported by 87.5% of patients and 96.3% of psychiatrists in cases where the patient refuses to eat with a threat to the life and health of the patient. Forced feeding should be used in any case of refusal to eat, according to 30.8% of psychiatrists and 20.3% of patients (χ^2 =4.205, P=0.04, OR=1.8, 95% CI, 1.0%-3.0%). Also, 12.5% of patients and 3.7% of psychiatrists pointed out the inadmissibility of force-feeding people with mental disorders in the psychiatric department (χ^2 =5.6438, P=0.018, OR=3.7, and 95% CI, 1.2%-12.63%).

On average, those with mental disorders preferred to have a somewhat more distant distance from psychiatric patients (3.4±1.9) than with "neighbors down the street" (Table 7). Psychiatrists chose a more distant one than with "city dwellers" (4.2±1.4). The first distance corresponds to a "geographical distance" of more than 1 km, and the second to more than 10 km. The possibility of close family and friendly relations with people with severe mental disorders was recognized by 40.6% of patients and 11.2% of psychiatrists (χ^2 =28.956, P=0.001, OR=5.4, and 95% CI, 2.7%-10.9%). Open relationships that allow formal social contact with neighbors and residents of the city suffering from mental disorders were considered possible by 23.2% of patients and 30.9% of psychiatrists.

In addition, 51.4% of psychiatrists and 25.5% of patients (x²=22.256 P=0.0005 OR=3.1 95% CI, 1.9%-5.1%) admit a distanced attitude towards psychiatric patients, the chance of a distanced attitude among psychiatrists is 3 times higher. 10.7% of patients and 6.5% of doctors rejected the desire to contact psychiatric patients - they were accepted only as "tourists in their country" and "would prefer not to see them in their country" ("isolation" and "rejection").

Discussion

Among the patients, those with severe social disadaptation prevailed, which is confirmed by previous studies of the contingent of patients in a psychiatric hospital [15]: More than half of them had severe mental disorders and disability due to mental disorders. Among non-disabled patients, nearly two-thirds were unemployed. The vast majority of patients completely or partially lacked criticism of their mental health, and they did not understand the essence of their disease due to pronounced self-stigmatization and anosognosia.

Recently, much attention has been paid to the study of psychiatrists' and patients' attitudes toward coercion [16-18], which may be related not only to the therapeutic function but also to the regulatory function of physical restraint. Our study showed that the majority of respondents in both groups considered it justified to force psychiatrists to treat people with mental disorders under any circumstances, which may be due to the dominant paternalistic model in psychiatry. However, the chance of a positive assessment of coercion by psychiatrists was four times higher than by patients. Only a small proportion of respondents in both groups spoke about the inadmissibility of the use of coercion and violence under any circumstances by psychiatrists, which is associated with its negative impact on the therapeutic process [19]. The opinion about coercion to treat people with mental disorders by nurses, among patients and psychiatrists was the same: More than half of the respondents in both groups recognized its admissibility. Moreover, among psychiatrists, the chance of a negative assessment of coercion by nurses was 2.6 times higher than patients, which is also confirmed by our other studies [20].

Psychiatrists had a positive attitude toward the possibility of personally applying physical restraint to an excited patient during voluntary hospitalization, which contradicts the requirements of article 30 of the law of the Russian Federation of 02.07.1992. No. 3185-1 [21]. Psychiatrists were 3.2 times more likely to be encouraged to indulge in physical restraint than patients, which is related to the role psychiatrists are charged with maintaining a safe environment in the psychiatric ward. The patients themselves also confirmed the physician's use of physical restraint and delegated safety concerns to psychiatrists and indirectly disclaimed responsibility for their actions due to anonymity. Respondents in both groups considered physical restraint to be a function of orderlies, which is confirmed by the results of other studies as a historical practice [22, 23]. At the same time, psychiatrists and patients supported the availability of special restraints in psychiatric hospitals. Psychiatrists were 2.2 times more likely to be encouraged to use physical restraint by nurses without a doctor's prescription than patients.

Among psychiatrists, the chance of a positive assessment of personal participation in the administration of a sedative injection was 4.2 times more than patients. Most patients endorsed helping other patients to prescribe sedatives to agitated patients, 3.7 times more than psychiatrists. This may be due to the lack of staffing in Russian psychiatric clinics; similar studies were not found.

According to our results, drug coercion ranks second among all types of coercion [24]. Patients were 3.7 times more likely to accept involuntary treatment of people with mental disorders than psychiatrists. Psychiatrists cited public danger as the only reason for involuntary treatment, with a chance more than 4.4 times higher than patients. The odds ratio showed that among patients, a negative assessment of compulsory treatment of persons with mental disorders occurs 2.7 times more often than among psychiatrists.

The majority of patients and psychiatrists approved drug control by nurses in the psychiatric ward, with psychiatrists being 3.1 times more likely to be positively assessed for control. Our results differ from other reports [25], which is due to the paternalistic model dominating Russian psychiatry. Naturally, patients were 3.2 times more likely than psychiatrists to talk about the inappropriateness of medication control, which is associated with a high prevalence of anosognosia among them. The opinions of both groups of respondents about forced feeding when patients in a psychiatric hospital refuse to eat were distributed in a similar manner.

Permissible social distance in relation to mentally ill patients among psychiatrists was significantly more distant than among patients, which is consistent with previous studies [26, 27], confirming the high level of stigmatization of patients by society and psychiatrists. The chance of having friendships and family relationships with this population was 5.4 times higher among the patients than doctors, indicating a lower level of stigma in patients. However, distanced attitudes among psychiatrists were 3.1 times more common than those with mental disorders.

Conclusion

The results showed that psychiatry is dominated by a paternalistic model of the doctor-patient relationship, which is fully supported by both parties. This model places psychiatrists in charge of making decisions concerning the health and lives of patients, as well as public safety and the interests of third parties. At the same time, the distanced attitude of psychiatrists and patients themselves toward people with mental disorders generally reflects the level of stigmatization of psychiatric patients in society and their self-stigmatization. The lack of a clear clinical and legal basis for limiting the rights and freedoms of psychiatric patients, together with the burden of personal responsibility of medical staff and their lack of legal security, leads to abuse, and overt or covert coercion. The high level of anosognosia in patients and

a certain ambivalent attitude towards hospitalization and treatment impose not only professional responsibility but also in some cases on the psychiatrist. The leading directions in the prevention of coercion and violence in a psychiatric hospital should increase the competence of the medical staff of psychiatric clinics in the field of medical law, the psycho-education of patients and their relatives, the use of a contract model in the organization of psychiatric care, and the responsibility of patients for their health, behavior, and treatment.

Ethical Considerations

Compliance with ethical guidelines

The performed procedures in the research involving human participants were in accordance with the ethical standards of the institutional and national study committee and the 1964 Helsinki Declaration and its later amendments or comparable ethical standards. The study protocol was approved by the Ethics Committee of the Medical Institute of the Belgorod State National Research University and complied with the provisions of the 1995 Helsinki Declaration, as amended and supplemented (Edinburgh, 2000)—protocol No. 12 dated February 6, 2018. Moreover, participating in the study was voluntary, and the study results are available to the study samples upon request.

Funding

This research did not receive any grant from funding agencies in the public, commercial, or non-profit sectors.

Conflict of interest

The author declared no conflict of interest.

Acknowledgments

The author acknowledge the patients and medical centers that contributed to this work.

References

[1] von Peter S, Aderhold V, Cubellis L, Bergström T, Stastny P, Seikkula J, et al. Open dialogue as a human rights-aligned approach. Frontiers in Psychiatry. 2019; 10:387. [DOI:10.3389/fpsyt.2019.00387] [PMID]

- [2] Babić D, Babić R, Vasilj I, Avdibegović E. Stigmatization of mentally Ill patients through media. Psychiatria Danubina. 2017; 29(Suppl 5):885-9. [PMID]
- [3] Shmukler AB, Gurovich IYa, Kostyuk GP. [Inpatient psychiatric care: deinstitutionalization and its consequences in Northern America and Europe (Russian)]. Social and Clinical Psychiatry. 2016; 26(4):97-109.
- [4] Seeman N, Tang S, Brown AD, Ing A. World survey of mental illness stigma. Journal of Affective Disorders. 2016; 190:115-21. [DOI:10.1016/j.jad.2015.10.011] [PMID]
- [5] Shvets KN, Khamskaya IS. [Factors of social disadaptation of patients with schizophrenia and approaches to psychosocial therapy and rehabilitation (review) (Russian)]. Research Results in Biomedicine. 2019; 5(2):72-85. [DOI:10.18413/2658-6533-2019-5-2-0-8]
- [6] Verbeke E, Vanheule S, Cauwe J, Truijens F, Froyen B. Coercion and power in psychiatry: A qualitative study with ex-patients. Social Science & Medicine. 2019; 223:89-96. [DOI:10.1016/j.socscimed.2019.01.031] [PMID]
- [7] Ebrahimi S, Salehi Dehno E. Attitudes of nurses and hospitalized patients about the rights of psychiatric clients. Iranian Journal of Psychiatry. 2018; 13(4):274-9. [PMID] [PMCID]
- [8] Raveesh BN, Lepping P. Restraint guidelines for mental health services in India. Indian Journal of Psychiatry. 2019; 61(Suppl 4):S698-705. [DOI:10.4103/psychiatry.IndianJPsychiatry_106_19] [PMID] [PMCID]
- [9] Sampogna G, Luciano M, Del Vecchio V, Pocai B, Palummo C, Fico G, et al. Perceived coercion among patients admitted in psychiatric wards: Italian results of the EUNOMIA study. Frontiers in Psychiatry. 2019; 10:316. [DOI:10.3389/fpsyt.2019.00316] [PMID]
- [10] Pollmächer T. [Autonomy focusing as guiding idea of minimally restrictive psychiatry (German)]. Der Nervenarzt. 2019; 90(7):669-74. [DOI:10.1007/s00115-019-0714-6] [PMID]
- [11] Gowda GS, Lepping P, Ray S, Noorthoorn E, Nanjegowda RB, Kumar CN, et al. Clinician attitude and perspective on the use of coercive measures in clinical practice from tertiary care mental health establishment A cross-sectional study. Indian Journal of Psychiatry. 2019; 61(2):151-5. [DOI:10.4103/psychiatry_Indian]Psychiatry_336_18] [PMID] [PMCID]
- [12] Sorokin MY, Lutova NB, Wied VD, Khobeysh MA, Makarevich OV. [Effects of psychopharmacotherapy and patients' adherence: Interdependence within the biopsychosocial paradigm (Russian)]. Research Results in Biomedicine. 2021; 7(2):202-14. [DOI:10.18413/2658-6533-2021-7-2-0-11]
- [13] Al Saif F, Al Shakhoori H, Nooh S, Jahrami H. Association between attitudes of stigma toward mental illness and attitudes toward adoption of evidence-based practice within health care providers in Bahrain. Plos One. 2019; 14(12):e0225738. [DOI:10.1371/journal.pone.0225738] [PMID]
- [14] Ruzhenkov VA, Ruzhenkova VV, Churnosova OI. [Results of studying social stigmatization of persons with mental disorders and suicidal behavior by senior students of medical faculty with the modified Bogardus Social Distance Scale (Russian)]. Bulletin of neurology, psychiatry and neurosurgery. 2017; 5-6. [Link]

- [15] Gurovich IYa, Shmukler AB, Kostyuk GP, Naryshkin AV. [Analysis of the patient population of a psychiatric hospital (According to the one-day assessment data)(Russian.)]. Social and Clinical Psychiatry. 2013; 2:5-14.
- [16] Tingleff EB, Bradley SK, Gildberg FA, Munksgaard G, Hounsgaard L. "Treat me with respect". A systematic review and thematic analysis of psychiatric patients' reported perceptions of the situations associated with the process of coercion. Journal of Psychiatric and Mental Health Nursings. 2017; 24(9-10):681-98. [DOI:10.1111/jpm.12410] [PMID]
- [17] Butterworth H, Wood L, Rowe S. Patients' and staff members' experiences of restrictive practices in acute mental health in-patient settings: Systematic review and thematic synthesis. BJPsych Open. 2022; 8(6):e178. [DOI:10.1192/bjo.2022.574] [PMID]
- [18] Chieze M, Clavien C, Kaiser S, Hurst S. Coercive measures in psychiatry: A review of ethical arguments. Frontiers in Psychiatry. 2021; 12:790886. [DOI:10.3389/fpsyt.2021.790886] [PMID]
- [19] Pérez-Toribio A, Moreno-Poyato AR, Roldán-Merino JF, Nash M. Spanish mental health nurses' experiences of mechanical restraint: A qualitative descriptive study. Journal of Psychiatric and Mental Health Nursing. 2022; 29(5):688-97. [DOI:10.1111/jpm.12860] [PMID]
- [20] Rzhevskaya NK, Merzlikina KS, Ruzhenkova VV, Retiunskii KY, Zamyatina II, Shiryaev OY. [Patients' attitude towards coercion and violence in providing stationary psychiatric aid: Ethical and preventive aspects (Russian)]. Research Results in Biomedicine. 2022; 8(1):106-16. [DOI:10.18413/2658-6533-2022-8-1-0-8]
- [21] ConsultantPlus. Russian federation law about psychiatric care and guarantees of citizens' rights when provided (Russian) [Internet]. 1992 [Updated 2023 October 1]. Available from: [Link]
- [22] Aguilera-Serrano C, Guzman-Parra J, Garcia-Sanchez JA, Moreno-Küstner B, Mayoral-Cleries F. Variables associated with the subjective experience of coercive measures in psychiatric inpatients: A systematic review. Canadian Journal of Psychiatry. 2018; 63(2):129-44. [DOI:10.1177/0706743717738491] [PMID]
- [23] Rabenschlag F, Cassidy C, Steinauer R. Nursing perspectives: Reflecting history and informal coercion in de-escalation strategies. Frontiers in Psychiatry. 2019; 10:231. [DOI:10.3389/fpsyt.2019.00231] [PMID]
- [24] Laukkanen E, Kuosmanen L, Selander T, Vehviläinen-Julkunen K. Seclusion, restraint, and involuntary medication in Finnish psychiatric care: A register study with root-level data. Nordic Journal of Psychiatry. 2020; 74(6):439-43. [DOI:10.1080/08039488.2020.1733658] [PMID]
- [25] Zinkler M, Waibel M. [Inpatient mental health care without mechanical restraint, seclusion or compulsory medication (German)]. Psychiatrische Praxis. 2019; 46(S01):S6-10. [DOI:10.1055/a-0785-6435] [PMID]
- [26] Bradstreet S, Dodd A, Jones S. Internalised stigma in mental health: An investigation of the role of attachment style. Psychiatry Research. 2018; 270:1001-9. [DOI:10.1016/j.psychres.2018.03.047] [PMID]

[27] Dubreucq J, Plasse J, Franck N. Self-stigma in serious mental illness: A systematic review of frequency, correlates, and consequences. Schizophrenia Bulletin. 2021; 47(5):1261-87. [DOI:10.1093/schbul/sbaa181] [PMID]